communications medicine

ARTICLE

https://doi.org/10.1038/s43856-02<u>2-00157-w</u>

OPEN

Check for updates

Using deep learning-based natural language processing to identify reasons for statin nonuse in patients with atherosclerotic cardiovascular disease

Ashish Sarraju^{1,5}, Jean Coquet ^{2,3,5}, Alban Zammit ^{2,3}, Antonia Chan², Summer Ngo¹, Tina Hernandez-Boussard ^{2,3,4,6} & Fatima Rodriguez ^{1,6⊠}

Abstract

Background Statins conclusively decrease mortality in atherosclerotic cardiovascular disease (ASCVD), the leading cause of death worldwide, and are strongly recommended by guidelines. However, real-world statin utilization and persistence are low, resulting in excess mortality. Identifying reasons for statin nonuse at scale across health systems is crucial to developing targeted interventions to improve statin use.

Methods We developed and validated deep learning-based natural language processing (NLP) approaches (Clinical Bidirectional Encoder Representations from Transformers [BERT]) to classify statin nonuse and reasons for statin nonuse using unstructured electronic health records (EHRs) from a diverse healthcare system.

Results We present data from a cohort of 56,530 ASCVD patients, among whom 21,508 (38%) lack guideline-directed statin prescriptions and statins listed as allergies in structured EHR portions. Of these 21,508 patients without prescriptions, only 3,929 (18%) have any discussion of statin use or nonuse in EHR documentation. The NLP classifiers identify statin nonuse with an area under the curve (AUC) of 0.94 (95% CI 0.93-0.96) and reasons for nonuse with a weighted-average AUC of 0.88 (95% CI 0.86-0.91) when evaluated against manual expert chart review in a held-out test set. Clinical BERT identifies key patient-level reasons (side-effects, patient preference) and clinician-level reasons (guideline-discordant practices) for statin nonuse, including differences by type of ASCVD and patient race/ ethnicity.

Conclusions Our deep learning NLP classifiers can identify crucial gaps in statin nonuse and reasons for nonuse in high-risk populations to support education, clinical decision support, and potential pathways for health systems to address ASCVD treatment gaps.

Plain language summary

Cardiovascular disease (CVD) is the leading cause of death worldwide. Statins are cholesterol-lowering drugs that are recommended by major guidelines to reduce the risk of heart attacks and death. However, despite being effective and generally well-tolerated, statins are concerningly underused, and reasons for such statin nonuse are not wellunderstood. Using artificial intelligence (AI) approaches that can analyze complex language data including written medical terminology, we identified statin nonuse and reasons for statin nonuse from clinical notes from electronic health records of CVD patients at a large health system. We found that nearly 2 in 5 patients lacked guideline-directed statin prescriptions. We then identified the key reasons for statin nonuse including patient (side-effects, personal preference) and clinician reasons (not practicing according to guidelines). Our Al approaches may help target interventions to address statin nonuse and thus, improve CVD treatment.

¹ Division of Cardiovascular Medicine and Cardiovascular Institute, Stanford University, Stanford, CA, USA. ² Department of Medicine, Stanford University, Stanford, CA, USA. ³ Department of Biomedical Data Science, Stanford University, Stanford, CA, USA. ⁴ Department of Surgery, Stanford University School of Medicine, Stanford, CA, USA. ⁵These authors contributed equally: Ashish Sarraju, Jean Coquet. ⁶These authors jointly supervised this work: Tina Hernandez-Boussard, Fatima Rodriguez. ¹²

A therosclerotic cardiovascular disease (ASCVD) remains the leading cause of morbidity and mortality worldwide despite the availability of numerous therapies¹. In ASCVD, statins conclusively reduce adverse events including myocardial infarctions, strokes, and mortality^{2–4}. Thus, by major guidelines including the American College of Cardiology/American Heart Association (ACC/AHA) multisociety guidelines, statins carry the strongest treatment recommendation in ASCVD in the absence of contraindications^{1,5}. However, statin use in ASCVD populations is alarmingly low, with discontinuation rates approaching 50% at one year following a myocardial infarction^{6,7}. Racial/ethnic minorities, women, and elderly patients are more likely to prematurely discontinue statins^{8–12}. Statin nonuse is independently associated with adverse outcomes including allcause mortality and represents an important public health gap¹³.

Reasons for statin nonuse can be multifactorial and complex, including patient, clinician, and system factors. These have typically been studied from surveys with potential selection bias and generalizability issues^{14–18}. Characterizing statin nonuse directly from electronic health records (EHRs) represents a unique opportunity to capture the local epidemiology of statin nonuse in a health system, which in turn may help develop targeted interventions to improve statin use. In clinical practice, reasons for statin nonuse are documented in narrative, free-text notes in the EHR, rather than in structured fields. Identifying them requires detailed characterization of large-scale unstructured EHR data¹¹.

Artificial intelligence (AI) technologies, including natural language processing (NLP), demonstrate promise in studying routine clinical data at scale, including for cardiovascular outcome prediction^{19–23}. We developed a deep learning-based NLP approach (Clinical Bidirectional Encoder Representations from Transformers [Clinical BERT]) to characterize reasons for statin nonuse directly from EHR data of a multiethnic, real-world, ASCVD cohort²⁴. The idea of this approach was to create a highly flexible deep learning model that can be incorporated into a clinical tool to accurately identify patients not meeting clinical guideline recommendations for statin use and then identify reasons for statin nonuse.

In this study of a multisite, multiethnic EHR-based health system, we found that approximately 40% of ASCVD patients lacked structured statin prescriptions. Using Clinical BERT, we accurately identified statin nonuse and key reasons for statin nonuse from unstructured clinical notes, including the prevalence of patient-level (side-effects, patient preference) and clinical-level (guideline-discordant practice) reasons. We observed differences in reasons for statin nonuse by patient race/ethnicity. By guiding targeted interventions to address statin nonuse, such a tool can bridge guideline-directed statin utilization gaps in diverse, realworld settings.

Methods

Study design. This retrospective study identified patients at Stanford Health Care Alliance (SHA) using EHR data from 1 January 2014 to 31 July 2019. SHA is an integrated health system, which includes an academic hospital (Stanford Health Care [SHC]), a community hospital (ValleyCare Hospital [ValleyCare]), and a community practice network (University Health-care Alliance [UHA]). The study was approved by the Stanford University Institutional Review Board (Protocol 47644). Informed consent was waived under exemption 4: research on existing data.

Patient cohort. The study included patients diagnosed with ASCVD between the ages of 18 and 89 years between 1 January 2014 and 31 July 2019 (Fig. 1). International Classification of

Diseases 9 and 10 (ICD 9 and 10) codes were used to identify the ASCVD diagnosis, which included coronary artery disease, cerebrovascular disease, peripheral arterial disease, and polyvascular disease (two or more ASCVD diagnoses; Supplementary Table 1). To ensure that patients received regular care within this healthcare system, patients were excluded if they did not have at least two encounters with an ASCVD diagnosis (Fig. 1). The first ASCVD diagnosis was considered the index diagnosis. RxNorm codes were used to identify statin medications (Supplementary Table 2). Patients with documented ASCVD diagnoses were first classified based on the presence or absence of statin prescriptions documented in structured EHR medication data at diagnosis date (Fig. 1). Statin prescriptions at diagnosis date or a new prescription within 1 month after diagnosis were included. In the case of a prescription that had no recorded end date, the prescription was considered active if it had started within 6 months prior to the index diagnosis date. Patients who lacked statin prescriptions were further classified by whether they had documented statin allergies in structured fields. Patients without documented allergies in structured fields were then further classified by the presence or absence of the term "statin" or the drug names in unstructured clinical notes (Supplementary Table 3). Types of clinical notes included were predefined and comprehensive in scope, spanning inpatient and outpatient care (Supplementary Data 1). Clinical notes dated up to 30 days after the index ASCVD diagnosis were included to allow the diagnosing providers up to 1 month after a new ASCVD diagnosis to document statin prescriptions, statin use, or statin nonuse. Encounter documentation is typically expected to be completed within 14 days in our health system. Patients who had any statin terms in clinical notes were included in the final cohort (deep learning NLP dataset, Fig. 1) for the development of the Clinical BERT models for the classification of statin nonuse and reasons for statin nonuse from unstructured clinical notes. We manually reviewed 20 random charts from patients who lacked statin terms to confirm the lack of statin use.

Patient characteristics. Clinical data were captured at the time of index ASCVD diagnosis for patients with and without structured statin prescriptions, including age, gender, race/ethnicity, history of smoking, number of hospitalizations in the year prior to diagnosis, and comorbidities based on ICD 9 and ICD 10 codes, including heart failure, atrial fibrillation, chronic kidney disease, and liver disease. Race/ethnicity was abstracted from the EHR. ASCVD diagnoses were categorized as coronary artery disease, cerebrovascular disease, peripheral arterial disease, or polyvascular disease (Supplementary Table 1). Laboratory studies, including total cholesterol, low-density lipoprotein cholesterol (LDL-C), and creatinine kinase (CK) or creatinine phosphokinase (CPK), were obtained within 6 months of initial diagnosis, and the value closest to the diagnosis date was used. The medical specialty of the encounter visit documenting the ASCVD diagnosis was captured. The patient's insurance payor was categorized as Medicare, Medicaid, or Private.

Primary outcomes. The primary outcome was statin nonuse based on structured and unstructured EHR data. The secondary outcome was the reason for statin nonuse classified into the following categories: muscle-based side-effects, other side-effects, perceived lipid control, patient preference, and nonspecific reasons. The categories of "muscle-based side-effects" and "other side-effects" included documentation of side-effects attributed to statin use as well as perceived contraindications to statin use attributed to pre-existing comorbidities (examples include underlying skeletal myopathy or advanced liver disease). The



Fig. 1 CONSORT-style diagram for cohort selection. Abbreviations: ASCVD atherosclerotic cardiovascular disease, SHA Stanford Health Care Alliance (consisting of an academic hospital, a community hospital, and a community practice clinic network), NLP natural language processing. "Statin allergy documented" refers to the documentation of a statin allergy in the structured "allergies" field of the EHR.

category of "perceived lipid control" was defined as documentation that statin use was avoided either based on available lipid levels or deferred in favor of obtaining future lipid tests to determine the need for statin therapy despite the presence of an ASCVD diagnosis. As contemporary ASCVD guidelines recommend the initiation of statin therapy as tolerated independent of baseline lipids, this was considered a guideline-discordant practice¹. The category of "patient preference" was defined as documentation of statin nonuse due to patient preference but without further documentation of a specific reason relevant to other categories. The category of "nonspecific" was defined as documentation of statin nonuse in clinical notes without a reason that may be included in other categories.

Natural language processing. For the deep learning NLP dataset, as outlined above, clinical notes were obtained from a 30 day period after the index ASCVD diagnosis. For each patient, all sentences which contained mentions of statins based on a predefined statin term dictionary based on drug names (Supplementary Table 3) were concatenated in one document for model training and evaluation. We employed Clinical BERT, a novel deep learning approach consisting of semi-supervised machine learning models pretrained on a large set of free texts which can be fine-tuned to a specific target task (including text classification) in a process called transfer learning²⁴. The Clinical BERT model was pretrained on notes from MIMIC III, a database containing EHRs from ICU patients at the Beth Israel Hospital in Boston, MA²⁴.

Manual annotation of reasons for statin nonuse. To create a manually annotated ground-truth dataset for NLP model training and evaluation, a sample of patients (N = 1742, 44%) was randomly selected from the deep learning NLP dataset. Four coauthors (AS, AC, SN, FR) manually annotated this dataset to determine whether clinical notes contained documentation of active statin use or nonuse, and subsequently extracted reasons for statin nonuse from notes according to categories. To assess reviewer concordance, the overlapping review was performed in a set of 50 patients and a set of 100 patients. Discrepancies between reviewers in these overlapping notes were resolved by repeat review by a clinician expert (AS or FR). Non-muscle side effect categories were defined more granularly prior to review (including liver side effects, gastrointestinal side effects, and neurological side effects), but were collapsed into a single category to comply with privacy regulations that prevent the reporting of groups with less than 10 participants. Categories were not considered mutually exclusive. The categories of reasons for statin nonuse were finalized upon completion of manual review.

NLP model to classify statin nonuse. We first developed the Clinical BERT model to determine statin use versus nonuse from clinical notes in the deep learning NLP dataset; that is, among

patients without statin prescriptions, we investigated clinical notes for documentation of statin use from external sources such as outside prescriptions. The fine-tuned model consisted of binary classification between patients with statin use documented in clinical notes (positive cases) and without documented statin use (negative cases).

NLP model to classify reasons for statin nonuse. In patients without documentation of statin use in clinical notes, we developed another model to determine documented reasons for statin nonuse, classified by the categories as previously described. We fine-tuned the Clinical BERT pretrained model to perform a multiclass classification with our dataset. For this task, we implemented a two-step pipeline. First, we trained five different models to classify one class (positive cases) against the four other classes (negative cases) in a one-versus-rest strategy²⁵. Second, we used a random forest model that took the probabilities of each reason from the first step as inputs to predict the final reason for nonuse. We also separately implemented a multilabel Clinical BERT classification model to predict the reason for statin nonuse.

NLP model training and evaluation. For both NLP models to classify statin nonuse and reasons for nonuse, we split the manually annotated patients into two data sets: a training set with 80% of patients and a test set with 20% of patients. We used 10fold cross-validation to validate the model and tune the hyperparameters (learning rate, number of epochs, strength of weight decay, and Adam's epsilon value). To evaluate the performance of models, we assessed precision, recall, Area Under Curve (AUC) score, and F1 score. Precision (or positive predictive value) measures the fraction of correct positive predictions divided by all documents predicted as positive. Recall (or true positive rate) measures the fraction of correct positive predictions divided by the number of results that should have been predicted as positive. AUC score corresponds to the probability that a classifier will rank a positive document higher than a negative one. Finally, F1 score is defined as the harmonic mean of precision and recall. For the multiclass classifiers for the reasons for statin nonuse, we reported the weighted macro-averages of these metrics. After training and evaluation, the binary NLP model and highest performing multiclass NLP model were applied across the full deep learning NLP dataset. The distribution of reasons for statin nonuse was stratified by type of ASCVD (coronary artery disease, cerebrovascular disease, peripheral arterial disease, and polyvascular disease) and race/ethnicity (Non-Hispanic White [NHW], Non-Hispanic Black, Hispanic, and Non-Hispanic Asian).

Statistical analysis. For comparisons of baseline characteristics between patients with and without statin prescription, we performed an unpaired *t* test for parametric data and chi-square/ Fisher's exact tests for categorical variables. Cohen's Kappa coefficients were calculated to assess the concordance of the manual annotations between overlapping reviewers. All statistical tests were two-sided with a threshold of $p \le 0.05$ for statistical significance. We calculated 95% confidence intervals for NLP model performance by performing a bootstrap resampling of 1000 iterations. We also compared unadjusted and adjusted Odds Ratios (OR) of receiving high-intensity statin prescriptions (versus other statin intensities) and of receiving any statin prescriptions (versus no prescriptions) through multivariable logistic regression. Analyses were performed using Python software, version 3.7, with transformers and scikit-learn packages.

Results

Cohort selection. There were 56,530 patients with documented ASCVD who had at least two separate visits to the healthcare system during the study period (Fig. 1). Of these patients, 22,296 (40%) were not prescribed statins of any intensity in EHR structured medication data (Fig. 1, Table 1). Patients without a statin prescription had a mean age of 65.5 ± 14.7 years, 45.1% were women, 58.4% were NHW, 5.8% were non-Hispanic Black (Black), 9.6% were Hispanic, and 13.6% were non-Hispanic Asian (Asian). Coronary artery disease was the most common ASCVD condition among patients without a statin prescription (34.0%). Patients without a statin prescription had a baseline LDL-C level of 107.9 ± 37.2 mg/dl (mean \pm standard deviation [SD]), compared with 90.2 ± 38.3 mg/dl in patients prescribed statins.

Of 22,296 patients without statin prescriptions, a total of 788 patients (3.5%) had documented statin allergies in the structured portion of the EHR. A manual review of 20 random patients with structured allergies confirmed the presence of allergies and lack of active statin use. Among the remaining 21,508 patients, a total of 17,579 (81.7%) had no mention of statin terms in clinical notes. The remaining 3929 patients who had any mention of statin terms in their clinical notes formed the deep learning NLP dataset for Clinical BERT model development and evaluation (Fig. 1).

Statin prescriptions. Across all 56,530 patients with ASCVD, women (compared with men), community hospital patients (compared with an academic hospital), and patients with cerebrovascular disease, peripheral arterial disease, or polyvascular disease (compared with coronary artery disease) were less likely to receive any statin prescriptions (Supplementary Table 4). Women (compared with men) and those with cerebrovascular disease or peripheral arterial disease (compared with coronary artery disease) were less likely to receive set by the set of the prescriptions (supplementary Table 5).

Manual annotation results. In the manually annotated dataset (Fig. 2), 742 (43%) were on statins per unstructured notes despite no prescription data. Among the 1000 (57%) statin nonusers, reasons for nonuse included perceived lipid control (17%, N = 174), muscle side-effects (14%, N = 139), and patient preference (15%, N = 153) (Fig. 2). Kappa coefficients for overlapping manual review of 50-patient and 100-patient subsets were 0.91 and 0.99, respectively, indicating strong reviewer agreement.

NLP model evaluation. In the held-out test set of the manually annotated dataset, the binary NLP model classified statin nonuse with an overall AUC of 0.94 (95% CI 0.93–0.96) (Table 2). Among statin nonusers, the two-step model classified reasons for statin nonuse with an overall weighted-average AUC of 0.88 (95% CI 0.86–0.91). The simple multilabel classification model classified reasons for statin nonuse with an overall AUC of 0.86 (95% CI 0.82–0.89).

NLP model application in the full cohort and by pre-specified subgroups. After NLP model development and evaluation, the binary statin nonuse model and the hurdle multiclass model for reasons for statin nonuse were applied to the full dataset of 3929 patients (Fig. 2). The models found that 1534 (39%) of these patients were statin users based on their clinical notes despite no documented statin prescriptions. Among the remaining 2395 statin nonusers, reasons for statin nonuse included muscle-based side-effects (16%), perceived lipid control (11%), and patient preference (13%; Fig. 2).

Age (years, mean ±SD) Female Race Non- Non- Hispa Non- Othe Provider location SHC UHA	Hispanic White Hispanic Black anic Hispanic Asian	68.6 ± 11.6 12179 (35.6%) 19086 (55.7%) 1687 (4.9%) 3167 (9.3%)	65.5 ± 14.7 10054 (45.1%) 13031 (58.4%) 1297 (5.8%%)	<0.001 <0.001
Female Race Non- Non- Hispa Non- Othe Provider location SHC UHA	Hispanic White Hispanic Black anic Hispanic Asian	12179 (35.6%) 19086 (55.7%) 1687 (4.9%) 3167 (9.3%)	10054 (45.1%) 13031 (58.4%) 1297 (5.8%%)	<0.001
Race Non- Non- Hispa Non- Othe Provider location SHC UHA	Hispanic White Hispanic Black anic Hispanic Asian	19086 (55.7%) 1687 (4.9%) 3167 (9.3%)	13031 (58.4%) 1297 (5.8%%)	
Non- Hispa Non- Othe Provider location SHC UHA	Hispanic Black anic Hispanic Asian	1687 (4.9%) 3167 (9.3%)	1297 (5.8%%)	<0.001
Hispa Non- Othe Provider location SHC UHA	anic Hispanic Asian	3167 (9.3%)		
Provider location SHC UHA	Hispanic Asian		2139 (9.6%)	
Provider location Othe SHC UHA	•	5553 (16.2%)	3028 (13.6%)	
Provider location SHC UHA	r	3123 (9.1%)	1656 (7.4%)	
UHA		20442 (59.7%)	11152 (50.0%)	<0.001
N7 II		12226 (35.7%)	9779 (43.9%)	
Valle	vCare	1552 (4.5%)	1191 (5.3%)	
ASCVD type Coro	narv arterv	20152 (58.9)	7577 (34.0%)	<0.001
Cere	brovascular	7155 (20.9)	5484 (24.6%)	
Perip	heral Arterial	3117 (9.1)	3072 (13.8%)	
Polyv	vascular	3810 (11.1%)	6163 (27.6%)	
Current smoking		2006 (5.9%)	925 (4.1%)	<0.001
Hospitalizations in prior 1 year	(N)	3000 (8.8%)	1831 (8.2%)	0.022
Insurance status Priva	te	6984 (20.4%)	5420 (24.3%)	< 0.001
Medi	icare	20097 (58.7%)	11696 (52.4%)	
Medi	icaid	2144 (6.3%)	1529 (6.8%)	
PCSK9 inhibitors		59 (0.2%)	68 (0.3%)	0.001
Ezetimibe		1697 (5.0%)	292 (1.3%)	< 0.001
Total cholesterol (mg/dL, mear	n ± SD)	167.2 ± 46.2	169.3 ± 39.8	0.698
LDL-cholesterol level at index (mg/dl , mean \pm SD)	90.2 ± 38.3	107.9 ± 37.2	<0.001
Chronic kidnev disease	8, 7, 7, 7, 7, 7, 7, 7, 7, 7, 7, 7, 7, 7,	5286 (15.4%)	2248 (10.0%)	< 0.001
Heart failure		6091 (17.8%)	2937 (13.2%)	< 0.001
Atrial fibrillation		5565 (16.3%)	3393 (15.2%)	0.003
Liver disease		1966 (5.7%)	1483 (6.7%)	< 0.001
Creatine kinase level (mean ± S	D)	252.3 ± 746.6	386.5 ± 1022	0.001
Encounter specialties Card	iology	10906 (31.8%)	7985 (35.8%)	< 0.001
Inter	nal medicine	3176 (9.2%)	1816 (8.1%)	
Fami	lv medicine	2263 (6.6%)	991 (4.4%)	
Vasc	ular surgerv	1486 (4.3%)	1093 (4.9%)	
Radio	nlogy	1181 (3.5%)	993 (4.4%)	
Neur	osurgerv	664 (19%)	1017 (4 6%)	
Prima	arv care	1106 (3.2%)	459 (2.0%)	
Fmer	gency medicine	977 (2.8%)	540 (2.4%)	
Neur	ology	802 (2.3%)	618 (2.8%)	
Anes	thesiology	791 (2.3%)	510 (2.3%)	
Othe	rs	10882 (31.8%)	6274 (28.1%)	

Table 1 Baseline characteristics of the ASCVD study cohort by the presence or absence of a statin prescription.

Reasons for statin nonuse were further stratified by type of ASCVD and by race/ethnicity (Table 3). Non-Hispanic black, Non-Hispanic Asian, and Hispanic patients had higher representation in the perceived lipid control group (a guideline-discordant practice) when compared with the other groups (p < 0.001). Patients with cerebrovascular disease had a higher representation in the perceived lipid control group when compared with other groups (p < 0.001). Excerpts denoting each reason for statin nonuse are outlined in Table 4.

Among patients without structured statin prescriptions, individuals who were statin users according to NLP were slightly more likely to be male and have coronary artery disease compared with statin nonusers according to NLP (Supplementary Table 6).

Discussion

Strong evidence conclusively supports the use of statins in patients with ASCVD to reduce cardiovascular morbidity and mortality. However, in a real-world, multiethnic cohort of ASCVD patients, approximately 40% lacked guidelineconcordant statin prescriptions. There was limited clinical documentation of statin nonuse overall. A deep learning NLP approach (Clinical BERT) reliably classified statin nonuse and reasons for statin nonuse from large-scale unstructured notes including patient-level (side-effects, patient preference) and clinical-level reasons (guideline-discordant practice).

The benefits of statin use in ASCVD are well-established¹; yet, there remain major gaps in statin utilization^{6,7}. Our study adds to this literature by highlighting a high proportion of ASCVD patients without documented statin prescriptions in a large health system, including disparities in statin prescriptions by gender, race/ethnicity, and practice setting (academic versus community-based). There is a strong need to develop interventions to bridge statin utilization gaps. Previous work studying reasons for statin nonuse has often relied on surveys that are potentially limited by selection bias, generalizability, and scalability^{15,16}. A prior EHR-based study addressed statin discontinuation among patients with existing prescriptions and used a rules-based approach to analyze notes¹¹. To our knowledge, our study is the first to employ a deep learning-based NLP approach to comprehensively track statin nonuse and reasons for statin nonuse across structured and unstructured EHR data. We capture statin nonuse in a multiethnic multisite health system in



Fig. 2 Training, internal validation, and application of a deep learning model (Clinical BERT) for natural language processing to identify statin nonuse and classify reasons for statin nonuse from unstructured clinical notes of patients with ASCVD. Abbreviations: BERT Bidirectional Encoder Representations from Transformers, NLP natural language processing.

Table 2 Performance of deep learning NLP models to characterize statin nonuse from unstructured clinical notes in perso	ons
with ASCVD.	

Task	Dataset	Precision*	Recall*	F1 score*	AUC*
Binary classification of statin use	10-fold cross-validation $(N = 1,393)$	0.88 (0.86-0.90)	0.82 (0.77-0.87)	0.85 (0.83-0.87)	0.94 (0.93-0.95)
	Test set (N = 349)	0.87 (0.82-0.91)	0.82 (0.76-0.88)	0.84 (0.81-0.88)	0.94 (0.93-0.96)
Two-step classifier* for statin nonuse reasons	10-fold cross-validation $(N = 800)$	0.63 (0.59-0.65)	0.62 (0.54-0.72)	0.62 (0.59-0.64)	0.84 (0.81-0.85)
	Test set (N = 200)	0.68 (0.63-0.75)	0.69 (0.60-0.79)	0.68 (0.62-0.75)	0.88 (0.86-0.91)
Multilabel classification of statin nonuse reasons (simple mutlilabel model)	10-fold cross-validation $(N = 800)$	0.60 (0.58-0.64)	0.61 (0.56-0.66)	0.59 (0.56-0.63)	0.85 (0.83-0.87)
	Test set (N $=$ 200)	0.64 (0.61-0.70)	0.66 (0.60-0.73)	0.64 (0.58-0.71)	0.86 (0.82-0.89)
[*] The two-step classifier represents the predicted probabi ASCVD atherosclerotic cardiovascular disease. NLP natur	lities of multiple classifiers (each re	ason for statin nonuse vers	us others) reconciled by a	Random Forest.	

detail, including documentation gaps, clinician- and patient-level reasons for nonuse, and differences by characteristics such as ASCVD type and race/ethnicity.

EHR-based studies of medication utilization often rely on structured data such as prescriptions. However, we found substantial discordance between structured and unstructured EHR documentation of statin use. Inconsistent documentation of external prescriptions and medical care may contribute to this finding²⁶ For example, we found clinical free-text documentation of active statin use in patients without statin prescriptions, suggesting that statin prescriptions were an unreliable surrogate for statin use in our cohort. Statin allergies in structured data were infrequently documented, but limiting side-effects were documented for patients without structured statin allergies, suggesting that using structured allergy data to infer statin intolerance would have been inaccurate. These results highlight the need for a standardized approach to document medication use in EHRs, a recognized issue for real-world data²⁷. Importantly, research efforts that study real-world medication utilization may need to mitigate the limitations of structured data by incorporating unstructured data at scale, as in our study.

Our study highlights poor clinical documentation of statin nonuse, with only 18% of patients without statin prescriptions demonstrating any mention of statins in clinical notes. These findings suggest important opportunities to improve statin nonuse documentation, for example, through smart text phrases or similarly integrated user-friendly tools that are guided by NLP. Ensuring adequate documentation is mandatory in learning health systems for quality improvement (QI) and research efforts that rely on large-scale records. A deep learning NLP pipeline like ours may be well-suited to study and bridge documentation gaps by leveraging unstructured EHR data²⁸.

Accurately characterizing reasons for nonuse is critical to closing statin utilization gaps. An approach like ours may enhance efforts to improve statin utilization by (1) identifying reasons for nonuse to guide targeted clinical decision support Table 3 NLP-identified reasons for statin nonuse in patients with ASCVD, stratified by type of ASCVD and race/ethnicity.

Cohort (N = 3929) Total number	Reason for nonuse					
	Side effect					
	Muscle O 385 40	Other	Nonspecific	Perceived lipid control 274	Patient preference 321	
		404	1011			
Stratified by type of ASCVD	(N, % by column)				
Coronary artery disease	224 (58.2)	233 (57.7)	472 (46.7)	123 (44.9)	175 (54.5)	<0.001
Peripheral artery disease	35 (9.1)	38 (9.4)	129 (12.8)	29 (10.6)	47 (14.6)	
Cerebrovascular disease	87 (22.6)	85 (21.0)	310 (30.7)	95 (34.7)	74 (23.1)	
Polyvascular disease	39 (10.1)	48 (11.9)	100 (9.9)	29 (10.6)	47 (14.6)	
Stratified by race/ethnicity ((N, % by column)					
Non-Hispanic White	244 (63.4)	273 (67.6)	554 (54.8)	132 (48.2)	214 (66.7)	<0.001
Non-Hispanic Black	13 (3.4)	25 (6.2)	62 (6.1)	19 (6.9)	13 (4.0)	
Hispanic	30 (7.8)	15 (3.7)	96 (9.5)	34 (12.4)	19 (5.9)	
Non-Hispanic Asian	40 (10.4)	46 (11.4)	96 (9.5)	34 (12.4)	19 (5.9)	

Table 4 Excerpts from clinical notes demonstrating the reasons for statin nonuse identified in this study.

Category	Note excerpt			
Muscle-based side-effects	"intolerant of low dose statins (started with high CK)"			
Other side-effects	"has been intolerant to 3 different statin drugs they cause diarrhea and nausea"			
Perceived lipid control	"Will also check lipid panel. If LDL < 100, no indication for statin at this time"; " LDL is well controlled"			
Patient Preference	"Declines statins"			
Nonspecific	"OK for no statin at this time"; "discuss statin next visit"			
ACCVD athereselentic cardiousceular disease. CK creatining kinase, LDL law density linguation chalacteral. NUP natural language processing				

tools, (2) identifying disparities by relevant clinical factors such as type of ASCVD, and (3) providing a pathway for EHR-driven statin utilization QI or research efforts. For example, patients with documented muscle-based side-effects could be flagged for further study using validated tools to confirm statin-related symptoms and to develop targeted interventions to re-challenge statins or address nocebo effects^{17,29}. Differences in statin nonuse by factors such as race/ethnicity or type of ASCVD may help guide equitable statin implementation strategies³⁰. Future work should prospectively explore the role of NLP-guided interventions to promote statin utilization.

Overall, the goal of our study was to build an accurate AIbased model to classify statin nonuse and reasons for nonuse among patients diagnosed with ASCVD. Using data derived from real-time EHRs, we trained our models to identify a patient that does not have any documentation of statin use for their ASCVD, and to identify reasons for statin nonuse. Such models are well-positioned to help overcome the challenges of real-world statin utilization by providing a better understanding of statin nonuse at scale. Such information will be a step forward towards improved statin use in routine care that is urgently needed to help reduce ASCVD disparities and gaps across diverse populations.

The study should be interpreted in the context of its limitations. Our cohort was a population from Northern California that may not reflect other ASCVD populations across the United States. However, our cohort is comprised of diverse practice sites including an academic hospital (SHC), a community hospital (ValleyCare), and a community practice network (UHA). We grouped non-muscle-based side-effects into a single category for the NLP model due to their low frequency and because we were unable to report outcomes with 10 or fewer participants due to HIPAA privacy regulations. It is possible that patients received additional care from outside health systems, which is a limitation of a single health system analysis. However, to mitigate the effects of care fragmentation and ensure regular care in our health system, we included patients with a new ASCVD diagnosis recorded in our system and at least 2 clinic visits associated with that diagnosis. Our EHR medication reconciliation section also includes medications prescribed outside of our health system and medications that are self-reported by patients. Future work should consider characterizing the frequency of more discrete categories of side-effects. We were unable to disaggregate racial/ethnic groups further or include socioeconomic information due to data limitations. Due to low overall documentation, the sample sizes of stratified groups (for example, muscle side-effects stratified by type of ASCVD) were small, and these results should therefore be interpreted with caution and considered hypothesis-generating. While Clinical BERT was pretrained on external notes and we report model performance from a heldout test set, external validation, as well as a prospective study of NLP-guided interventions, is needed to ensure generalizability, outcome benefit, and cost-effectiveness prior to wide deployment.

In conclusion, in multiethnic ASCVD patients in a multisite health system, we observed suboptimal statin prescription rates and limited clinical documentation of statin nonuse. A deep learning NLP approach (Clinical BERT) reliably identified statin nonuse and reasons for nonuse from unstructured EHRs, including patient-level factors (side-effects, patient preference) and clinician-level factors (guideline-discordant practice). Advanced NLP approaches may help to learn health systems to leverage clinical documentation and characterize reasons for statin nonuse at scale from EHRs, thus potentially providing a pathway to address important ASCVD treatment gaps. **Reporting summary**. Further information on research design is available in the Nature Research Reporting Summary linked to this article.

Data availability

The datasets analyzed during the current study are not publicly available due to reasonable privacy and security concerns. The underlying EHR data are not easily redistributable to researchers other than those engaged in the Institutional Review Board-approved research collaborations in the current project. The corresponding author may be contacted for access to EHR data for an IRB-approved collaboration.

Code availability

The code can be accessed at https://doi.org/10.25740/gn923yy3398³¹.

Received: 12 October 2021; Accepted: 29 June 2022; Published online: 15 July 2022

References

- Grundy, S. M. et al. 2018 AHA/ACC/AACVPR/AAPA/ABC/ACPM/ADA/ AGS/APhA/ASPC/NLA/PCNA Guideline on the management of blood cholesterol: executive summary: a report of the american college of cardiology/ american heart association task force on clinical practice guidelines. J. Am. Coll. Cardiol. 73, 3168–3209 (2019).
- Cannon, C. P. et al. Ezetimibe added to statin therapy after acute coronary syndromes. N. Engl. J. Med. 372, 2387–2397 (2015).
- LaRosa, J. C. et al. Intensive lipid lowering with atorvastatin in patients with stable coronary disease. N. Engl. J. Med. 352, 1425–1435 (2005).
- Goff, D. C. Jr. et al. 2013 ACC/AHA guideline on the assessment of cardiovascular risk: a report of the American College of Cardiology/American Heart Association Task Force on Practice Guidelines. J. Am. Coll. Cardiol. 63, 2935–2959 (2014).
- Stone, N. J. et al. 2013 ACC/AHA guideline on the treatment of blood cholesterol to reduce atherosclerotic cardiovascular risk in adults: a report of the American College of Cardiology/American Heart Association Task Force on Practice Guidelines. J. Am. Coll. Cardiol. 63, 2889–2934 (2014).
- McGinnis, B. et al. Factors related to adherence to statin therapy. Ann. Pharmacother. 41, 1805–1811 (2007).
- Brookhart, M. A. et al. Physician follow-up and provider continuity are associated with long-term medication adherence: a study of the dynamics of statin use. Arch. Intern. Med. 167, 847–852 (2007).
- Vinogradova, Y., Coupland, C., Brindle, P. & Hippisley-Cox, J. Discontinuation and restarting in patients on statin treatment: prospective open cohort study using a primary care database. *BMJ* 353, i3305 (2016).
- Maddox, T. M. et al. Implications of the 2013 ACC/AHA cholesterol guidelines for adults in contemporary cardiovascular practice: insights from the NCDR PINNACLE registry. J. Am. Coll. Cardiol. 64, 2183–2192 (2014).
 Rodriguez, F. et al. Use of high-intensity statins for patients with
- atherosclerotic cardiovascular disease in the Veterans Affairs Health System: practice impact of the new cholesterol guidelines. *Am. Heart J.* **182**, 97–102 (2016).
- Zhang, H. et al. Discontinuation of statins in routine care settings: a cohort study. Ann. Intern. Med. 158, 526–534 (2013).
- 12. Booth, J. N., 3rd et al. Statin discontinuation, reinitiation, and persistence patterns among medicare beneficiaries after myocardial infarction: a cohort study. *Circ. Cardiovasc. Qual. Outcomes* **10**, e003626 (2017).
- Rodriguez, F. et al. Association of statin adherence with mortality in patients with atherosclerotic cardiovascular disease. JAMA Cardiol. 4, 206–213 (2019).
- Hirsh, B. J., Smilowitz, N. R., Rosenson, R. S., Fuster, V. & Sperling, L. S. Utilization of and adherence to guideline-recommended lipid-lowering therapy after acute coronary syndrome: opportunities for improvement. *J. Am. Coll. Cardiol.* 66, 184–192 (2015).
- Jacobson, T. A. et al. The STatin adverse treatment experience survey: experience of patients reporting side effects of statin therapy. *J. Clin. Lipidol.* 13, 415–424 (2019).
- Cohen, J. D., Brinton, E. A., Ito, M. K. & Jacobson, T. A. Understanding Statin Use in America and Gaps in Patient Education (USAGE): an internet-based survey of 10,138 current and former statin users. J. Clin. Lipidol. 6, 208–215 (2012).
- Wood, F. A. et al. N-of-1 trial of a statin, placebo, or no treatment to assess side effects. N. Engl. J. Med. 383, 2182–2184 (2020).

- Herrett, E. et al. Statin treatment and muscle symptoms: series of randomised, placebo controlled n-of-1 trials. *BMJ* 372, n135 (2021).
- Maddox, T. M. & Matheny, M. A. Natural language processing and the promise of big data: small step forward, but many miles to go. *Circ. Cardiovasc. Qual. Outcomes* 8, 463–465 (2015).
- 20. Deo, R. C. Machine learning in medicine. Circulation 132, 1920–1930 (2015).
- Ward, A. et al. Machine learning and atherosclerotic cardiovascular disease risk prediction in a multi-ethnic population. *NPJ Digit. Med.* 3, 125 (2020).
- Bozkurt, S. et al. Is it possible to automatically assess pretreatment digital rectal examination documentation using natural language processing? A single-centre retrospective study. *BMJ Open* 9, e027182 (2019).
- 23. Banda, J. M. et al. Finding missed cases of familial hypercholesterolemia in health systems using machine learning. *NPJ Digit. Med.* **2**, 23 (2019).
- Alsentzer, E. et al. Publicly Available Clinical BERT Embeddings. arXiv 1904, 03323 https://ui.adsabs.harvard.edu/abs/2019arXiv190403323A (2019).
- Galar, M., Fernández, A., Barrenechea, E., Bustince, H. & Herrera, F. An overview of ensemble methods for binary classifiers in multi-class problems: experimental study on one-vs-one and one-vs-all schemes. *Pattern Recognit.* 44, 1761–1776 (2011).
- Rosenbloom, S. T. et al. Data from clinical notes: a perspective on the tension between structure and flexible documentation. J. Am. Med. Inform. Assoc. 18, 181–186 (2011).
- Hernandez-Boussard, T., Monda, K. L., Crespo, B. C. & Riskin, D. Real world evidence in cardiovascular medicine: ensuring data validity in electronic health record-based studies. J. Am. Med. Inform. Assoc. 26, 1189–1194 (2019).
- Quer, G., Arnaout, R., Henne, M. & Arnaout, R. Machine learning and the future of cardiovascular care: JACC state-of-the-art review. J. Am. Coll. Cardiol. 77, 300–313 (2021).
- Rosenson, R. S. et al. An assessment by the statin muscle safety task force: 2014 update. J. Clin. Lipidol. 8, S58–S71 (2014).
- Gamboa, C. M., Colantonio, L. D., Brown, T. M., Carson, A. P. & Safford, M. M. Race-sex differences in statin use and low-density lipoprotein cholesterol control among people with diabetes mellitus in the reasons for geographic and racial differences in stroke study. J. Am. Heart Assoc. 6, e004264 (2017).
- Sarraju, A. and Coquet, J. (2021). Using Deep Learning-based Natural Language Processing to Identify Reasons for Statin Nonuse in Patients with Atherosclerotic Cardiovascular Disease. Stanford Digital Repository. Available at https://purl.stanford.edu/gn923yy3398.

Acknowledgements

Dr. Ashish Sarraju received support from the American Heart Association (Grant 20SFRN35360178). Dr. Fatima Rodriguez received support from the National Heart, Lung, and Blood Institute, National Institutes of Health (1K01HL144607), and the American Heart Association/Robert Wood Johnson Harold Amos Medical Faculty Development Program.

Author contributions

A.S.: study design, data analysis and interpretation, manuscript preparation, and editing; J.C.: study design, data analysis and interpretation, manuscript preparation, and editing; A.Z.: data analysis and interpretation, manuscript preparation, and editing; A.C.: data analysis and interpretation, manuscript editing; S.N.: data analysis and interpretation, manuscript editing; T.H.B.: study conceptualization and design, data analysis and interpretation, manuscript editing, supervision; F.R.: study conceptualization and design, data analysis and interpretation, manuscript editing, and supervision.

Competing interests

F. Rodriguez has received consulting fees from Novartis, NovoNordisk, and HealthPals. T. Hernandez-Boussard has received consulting fees from Verantos and Institute for Systems Biology. The remaining authors have no competing interests to declare.

Additional information

Supplementary information The online version contains supplementary material available at https://doi.org/10.1038/s43856-022-00157-w.

Correspondence and requests for materials should be addressed to Fatima Rodriguez.

Peer review information *Communications Medicine* thanks the anonymous reviewers for their contribution to the peer review of this work.

Reprints and permission information is available at http://www.nature.com/reprints

Publisher's note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons license, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons license and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this license, visit http://creativecommons.org/ licenses/by/4.0/.

© The Author(s) 2022